Author's response to reviews

Title: How does healthcare worker handwashing behaviour impact upon the transmission of MRSA between patients? : an analysis using a Monte Carlo model

Authors:

Clive B Beggs (C.B.Beggs@bradford.ac.uk)
Simon J Shepherd (S.J.Shepherd@bradford.ac.uk)
Kevin G Kerr (kevin.kerr@hdft.nhs.uk)

Version: 4 Date: 5 May 2009

Author's response to reviews:

Major revisions

1. Figures in text have been deleted. Figure legends section added.
2. Tables removed to a new section after the References
3. Full details for reference now provided
4. Figures 1a,b,c have now been renumbered 1,2,3

Minor revisions

1. Author list amended
2. Email addressess given on the title page
3. Introduction changed to Background
4. Figures cropped
5. Figures cited individually
6. Check for typos undertaken